Lawrence Berkeley National Laboratory

LBL Publications

Title

Thirty years of North American wind energy acceptance research: What have we learned?

Permalink https://escholarship.org/uc/item/3747t3q4

Authors

Rand, Joseph Hoen, Ben

Publication Date

2017-07-01

DOI

10.1016/j.erss.2017.05.019

Peer reviewed

Thirty years of North American wind energy acceptance research: What have we learned?

Authors:

Joseph Rand, Ben Hoen

Energy Analysis and Environmental Impacts Division Lawrence Berkeley National Laboratory

Electricity Markets and Policy Group

June 2017

This is a pre-print of an article accepted for publication in *Energy Research and Social Science*. DOI: <u>https://doi.org/10.1016/j.erss.2017.05.019</u>



This work was supported by the Wind Energy Technologies Office within the Office of Energy Efficiency and Renewable Energy of the U.S. Department of Energy, under Contract No. DE-AC02-05CH11231.

Disclaimer

This document was prepared as an account of work sponsored by the United States Government. While this document is believed to contain correct information, neither the United States Government nor any agency thereof, nor The Regents of the University of California, nor any of their employees, makes any warranty, express or implied, or assumes any legal responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by its trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof, or The Regents of the University of California. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof, or The Regents of the University of California.

Ernest Orlando Lawrence Berkeley National Laboratory is an equal opportunity employer.

Acknowledgements

This work was funded by the Wind Energy Technologies Office within the Office of Energy Efficiency and Renewable Energy of the U.S. Department of Energy, under Contract No. DE-AC02-05CH11231. The authors specifically thank Patrick Gilman, Jocelyn Brown-Saracino, and Jose Zayas from the DOE Wind Energy Technologies Office for supporting this research

The authors would also like to thank Jeremy Firestone (University of Delaware), Ryan Wiser (Lawrence Berkeley National Laboratory), Eric Lantz (National Renewable Energy Laboratory), Gundula Hübner (Martin Luther University), and Jarett Zuboy for their contributions and suggestions to improve this review.

Table of Contents

Acl	knowle	dgementsi	
Tał	ole of C	Contentsii	
Abs	stract.	iii	
1.	Introduction1		
	1.1	Background and Motivation1	
	1.2	Justification for North American Focus	
2.	Metho	od of Literature Review	
	2.1	Selection of Publications to Review	
	2.2	Data Collection, Coding, and Qualitative Analysis of Papers	
3.	. A Brief History of North American Wind Energy Acceptance Research		
4.	. Limitations of Previous North American Wind Acceptance Research		
5.	Overarching Aspects and Explanatory Variables in North American Literature		
	5.1	Socioeconomic Aspects	
	5.2	Sound Annoyance and Health Risk Perceptions12	
	5.3	Visual/Landscape Aspects, Annoyance, and Place Attachment13	
	5.4	Environmental Concerns and Attitudes15	
	5.5	Perceptions of Planning Process, Fairness, and Trust16	
	5.6	Distance from Turbines (Proximity Hypothesis)18	
	5.7	Other Proposed Correlates of Acceptance	
6.	Lessons learned in 30 years of wind acceptance studies19		
7.	Knowledge gaps after 30 years: Areas for future research21		
8.			
Ref	erence	es	
Ap	pendix	A	
	Table 1: Explanatory variables in N.A. wind acceptance literature & research suggestions30		
	Table 2: Research approaches and methods in North American wind acceptance literature33		

Abstract

Thirty years of North American research on public acceptance of wind energy has produced important insights, yet knowledge gaps remain. This review synthesizes the literature, revealing the following lessons learned. (1) North American support for wind has been consistently high. (2) The NIMBY explanation for resistance to wind development is invalid. (3) Socioeconomic impacts of wind development are strongly tied to acceptance. (4) Sound and visual impacts of wind facilities are strongly tied to annoyance and opposition, and ignoring these concerns can exacerbate conflict. (5) Environmental concerns matter, though less than other factors, and these concerns can both help and hinder wind development. (6) Issues of fairness, participation, and trust during the development process influence acceptance. (7) Distance from turbines affects other explanatory variables, but alone its influence is unclear. (8) Viewing opposition as something to be overcome prevents meaningful understandings and implementation of best practices. (9) Implementation of research findings into practice has been limited. The paper also identifies areas for future research on wind acceptance. With continued research efforts and a commitment toward implementing research findings into developer and policymaker practice, conflict and perceived injustices around proposed and existing wind energy facilities might be significantly lessened.

Keywords:

Wind energy; social acceptance; support and opposition; attitudes

1. Introduction

1.1 Background and Motivation

Over the last 30 years, wind energy in North America has evolved from a fringe, isolated, experimental concept into a mainstream and viable source of electricity, meeting about 5% of U.S. electricity demand (6% in Canada) and representing the largest source of new electric capacity additions in many recent years (CanWEA, 2016; Wiser & Bolinger, 2016). Wind energy is widely seen as an abundant electricity source with the potential to provide a wide range of environmental and social benefits (Intergovernmental Panel on Climate Change (IPCC), 2011). State/provincial-level mandates, federal incentives, declining wind energy costs, and relatively favorable economics have spurred the aggressive North American wind deployment of the past 10–15 years (Wiser & Bolinger, 2016).

This rapid growth in wind energy deployment will likely continue. In the United States, for example, recent market analysis suggests that annual wind power capacity additions are expected to continue rapidly in the coming five years (Wiser & Bolinger, 2016, p. 1) driven by expected lower prices (Wiser et al., 2016). Meanwhile, the U.S. Department of Energy's recent *Wind Vision Report*, which outlines pathways for wind energy to provide up to 35% of the nation's electrical demand by 2050, suggests that the "low hanging fruit" wind sites (those that have good wind resources and are close to loads and transmission, yet far from communities) have largely been developed, implying that future wind development likely will happen increasingly near communities. As such, the report underlines the need for a better understanding of the drivers of wind facility acceptance among affected communities (USDOE, 2015). This recommendation echoes the calls of numerous social scientists, who have suggested that successful implementation of U.S. wind projects relies on a deeper understanding of local stakeholders (e.g., Petrova, 2013).

Multiple facets of acceptance can impact the deployment of renewable energy projects. Wüstenhagen et al. (2007) point to three dimensions: *Sociopolitical acceptance* (acceptance of policymakers and key stakeholders), *market acceptance* (acceptance of investors and consumers), and *community acceptance* (pertaining to procedural justice, distributional justice, and trust). However, as Sovacool (2009, p. 4511) points out, these social, technical, economic, and political dimensions of acceptance all influence each other in an integrated, "pernicious tangle." For example, community acceptance of wind energy can affect market acceptance and vice versa. Indeed, this has been the case when local opposition has delayed or derailed proposed wind projects (Corscadden et al., 2012; Fast, 2015; Shaw et al., 2015). For years, debates around wind energy acceptance in North America focused on sociopolitical and market acceptance, pertaining largely to technological innovation, economic incentives, and impacts on the operations and resiliency of the electric grid, with less attention paid to societal impacts (Lantz & Flowers, 2011; Phadke, 2010). However, the rapid growth of North American wind energy has increased the footprint of wind developments, increasing local conflicts and bringing the issue of community acceptance to the forefront (Lantz & Flowers, 2011).

Despite broad public support for wind energy in general, local wind developments have been challenged by vocal opposition within host communities (Bidwell, 2013; Bohn & Lant, 2009; Lantz &

Flowers, 2011). In the early days of U.S. wind power, opposition and negative attitudes dismayed the industry, who expected local acceptance to be consistent with the favorable opinions toward wind power generally (Pasqualetti, 2001). Ever since opposition and negative attitudes emerged around some of the earliest experimental wind farms in California, researchers have tried to understand and characterize wind energy acceptance (see Bosley & Bosley, 1988; Pasqualetti & Butler, 1987; Thayer & Freeman, 1987). Community acceptance is now widely perceived by wind energy practitioners as a significant barrier to deployment of renewable energy (Lantz & Flowers, 2011).

Research interest in the public acceptance of wind energy has surged along with surging wind energy growth in North America. After three decades (1987–2017) of North American scholarship in this field, what have we learned, how can these lessons be applied, and what aspects should researchers focus on next?

This review synthesizes the findings from more than 100 papers on wind energy acceptance published over the past 30 years, with a specific focus on the North American set of literature.

1.2 Justification for North American Focus

The North American¹ body of wind energy acceptance literature merits this its own review, distinct from the literature of Europe (which represents a vast body of literature; see, e.g. Ellis & Ferraro (2016)) and other regions. Social acceptance is highly context dependent, and Canada and the United States share many aspects of culture, national energy policy and economics, population density, land use policy, wind energy development processes, and wind project ownership models that are distinct from Europe and the rest of the world.

North America currently represents nearly 1/5 of global installed wind energy capacity (Global Wind Energy Council (GWEC), 2017), and the growth rate of that capacity over the past 15 years is markedly faster in North America than in Europe. Since 2002, installed capacity has increased nearly twenty-fold in North America, compared to a seven-fold increase in Europe (AWEA, 2003; Global Wind Energy Council (GWEC), 2017). This rapid growth of land-based wind energy in North America may result in amplified impacts to host communities that should be studied independently from the European context, where onshore development has been slower in recent years. Similarly, the sluggish growth of offshore wind energy in North America may also indicate economic, environmental, cultural, and visual concerns that are unique from European experience and worthy of independent study. The first offshore wind farm in North America, a 30-Megawatt project, was installed in 2016, while over 12.6 Gigawatts had been installed in Europe by the end of the same year (Global Wind Energy Council (GWEC), 2017). The density of population in Europe, coupled with the density of land-based wind turbines, places a greater proportion of the European population in closer proximity to turbines compared to North America, which may conceivably influence aspects of acceptance.

North America is largely electricity independent with domestic reserves of coal, natural gas, uranium,

¹ Very few studies from Mexico were found when preparing this review. The vast majority of papers reviewed herein focus on Canada and the United States.

hydropower, and other energy resources, whereas Europe is a net importer of fuels for electricity generation. These relatively cheap, domestic energy resources create steeper market competition and different economic conditions for wind deployment in North America compared to Europe. The European Union's Emissions Trading Scheme (EU ETS) represents an EU-wide, comprehensive climate policy; quite distinct from the state- or province-level energy and climate policies previously seen in North America. These are important differences in aspects of market and sociopolitical acceptance (both of which influence community acceptance, as described above) between North America and Europe.

Finally, community ownership and investment models are much less common in North America than in Europe (Bolinger, 2005; Ferguson-Martin & Hill, 2011; Sovacool & Ratan, 2012). In the United States, federal production incentives for wind energy have required a significant tax liability, tipping the scales toward large private developers of wind projects (Bolinger, 2005). Some studies have demonstrated that community ownership is correlated to higher support and more positive attitudes toward wind energy in Europe and other regions (Krohn & Damborg, 1999; Maruyama et al., 2007; Petrova, 2013; Warren & McFadyen, 2010). One may reasonably expect some differences in perceptions and acceptance of wind energy in North America in relation to the low level of community ownership in the region.

Despite this explicit geographic focus, a number of European papers are included in this review where those papers either introduce a novel concept or explanation that has since been studied in the North American context (e.g., place attachment theory), or point out broad aspects of the field of study of wind energy acceptance such as biases and limitations in previous research (see Section 4).

The following section outlines the methodological approach to reviewing the North American body of wind acceptance research. Section 3 provides a brief history of North American wind acceptance research to frame the papers discussed in this review. Section 4 discusses the limitations of previous research that have hindered meaningful understandings. Section 5 examines in detail the dominant explanations and overarching aspects of wind energy acceptance in the North American context. Section 6 provides a high-level summary of lessons learned from 30 years of research. Finally, Section 7 identifies gaps and areas where future research on the public acceptance of North American wind energy should focus.

2. Method of Literature Review

2.1 Selection of Publications to Review

The goal of this review was to broadly represent the body of research on wind energy acceptance undertaken in North America. Papers were initially solicited from a panel of five researchers in the field of wind energy acceptance in October, 2014. Additional papers were selected from internet searches using Google Scholar and Scopus² and from those cited in the papers (i.e. "snowballing"). Internet

² See https://www.elsevier.com/solutions/scopus

searches focused on the most recent studies, 2014 and later, which were less well represented in the panel solicitation.

The solicitation and searches focused on papers written by authors with North American affiliations and/or research pertaining to North American wind facilities. Although the United States and Canada are well represented in this review, very few papers from Mexico were found in the literature. The review focuses on peer-reviewed journal articles, but some books are included as well as some grey literature, primarily in the form of government-sponsored reports. There was no restriction placed on the date range of the papers selected for review. The earliest relevant North American studies were published in 1987, with a clear growing trend in publications per year in this field through 2016 (see Figure 1). Selected papers represent a broad range of published years in order to capture the evolution in this body of literature over time.

Papers were selected to represent a broad array of themes and variables that are examined by their authors, as well as diversity in the research approach and methods employed. The array of acceptance-influencing variables examined in these papers is outlined in Table 1 of the Appendix. The research methods and approaches utilized by the studies examined in this review are outlined in the Appendix Table 2. These tables not only summarize the literature reviewed in this paper, but also serve to clearly illustrate particular explanatory variables and research methods that have been applied in the North American literature, thus elucidating gaps.



Figure 1: North American wind energy acceptance papers, 1987-2016. Data source: Scopus³

³ This search was conducted using the Scopus database on April 25, 2017 using the following search string: TITLE-ABS-KEY ("wind energy" AND (public OR acceptance)) AND (LIMIT-TO (AFFILCOUNTRY , "United States ") OR LIMIT-TO (AFFILCOUNTRY , "Canada ") OR LIMIT-TO (AFFILCOUNTRY , "Mexico "))

2.2 Data Collection, Coding, and Qualitative Analysis of Papers

As each paper was read, it was entered into a detailed summary matrix to catalog the year, research questions, methods, analysis techniques, location, wind energy statistics (if applicable), sample size, explanatory variables examined, major conclusions, and additional research recommendations of each study.

These data became the framework for qualitative content analysis of the papers, which relied on interpreting the results, discussions, conclusions, and significant explanatory variables identified in the papers. This qualitative analysis led to the creation of the major themes detailed in Section 5, as well as the lessons learned outlined in Section 6, and the recommendations for further research discussed in Section 7 of this paper.

For quantitative studies, themes were identified by explanatory variables that were found to have a statistically significant effect on attitudes, support, or opposition. However, in some cases, a major contribution of particular papers was to *not* find a statistically significant effect of certain variables. Those were also included in the broad themes of this review. Similarly with qualitative and mixed-methods studies, the themes emerging from interviews, focus groups, and content analyses were categorized among the dominant themes identified in the broader body of literature. Table 1 of Appendix A summarizes these major explanatory themes that emerged from the literature, and identifies which papers from this review address each theme.

From the body of papers selected for this review, six overarching themes explaining attitudes and/or support and opposition toward wind energy emerged. Within each theme, a number of sub-themes also existed. These explanatory themes (along with their sub-themes) are analyzed and summarized in detail in Section 5 of this paper. The results and major conclusions of each paper are examined and contrasted to each other within each theme or explanatory group. The purpose of this analysis was to identify those explanatory variables with either broad agreement or considerable disagreement among the studies reviewed. Agreement among numerous studies would represent lessons learned (Section 6), where considerable disagreement (or a lack of research) may indicate a need for further research and/or suggestions for new methodologies (Section 7).

3. A Brief History of North American Wind Energy Acceptance Research

Academic research seeking to understand the acceptance of North American wind energy began in earnest shortly after some of the first experimental wind farms were installed in California in the 1980s. Surveys by Pasqualetti and Butler (1987) and Thayer and Freeman (1987) found a range of opinions among nearby residents, with negative attitudes most closely correlated with feelings of aesthetic degradation and frustration about non-functioning (i.e., non-spinning) turbines. Drivers of negative attitudes cited by Bosley & Bosley (1988) include a lack of knowledge about wind energy's "maturity" among opponents along with a failure on the part of the wind industry to communicate properly with affected parties. Gipe (1995) dedicated a full chapter to turbine aesthetics and community acceptance in his book, *Wind Energy Comes of Age.* Gipe specifically addresses aesthetic design, turbine reliability, the pace and process of planning and development, the distribution of costs and benefits (i.e., compensation), and community ownership models as potential determinants of acceptance (Gipe, 1995). These early studies led the way for three decades of scholarship in the field of wind energy acceptance.

According to this literature, wind energy has been viewed favorably throughout North America over the past 30 years, with 70%–90% of those surveyed approving of wind energy generally (Bidwell, 2013; Klick & Smith, 2010; Vyas & Hurst, 2013) and that approval remaining high over time (Vyas & Hurst, 2013). Support also has been high among residents of communities where wind projects have been proposed but not yet built (Firestone et al., 2009; Firestone et al., 2012a; Mulvaney et al., 2013b). In studies of people near existing wind facilities, 70%–90% of respondents express positive or neutral attitudes (Baxter et al., 2013; Fergen & Jacquet, 2016; Mulvaney et al., 2013b; Pasqualetti & Butler, 1987; Petrova, 2014; Slattery et al., 2012). Nonetheless, the 10%–30% of individuals who do *not* support proposed wind developments or hold negative attitudes toward existing facilities—as well as the factors influencing those positions—are of strong interest to the research community. Many researchers also seek to identify ways to minimize negative factors from existing and future wind developments.

In any case, opposition to wind development and negative attitudes toward existing wind installations are normal, and they likely will exist as long as wind facilities exist. The same can be said about other large construction projects. Transmission lines, landfills, and parks are among the types of projects that have been opposed (Gipe, 1995), and some people even protested the location and appearance of the Statue of Liberty (Petrova, 2013). Every component of our current electrical system was the product of social negotiation and compromise, including over "17,000 conventional generators, 250,000 miles of high voltage transmission lines, thousands of substations," and much more (Sovacool, 2009, p. 4502).

The rich body of research on wind energy acceptance spans myriad disciplines, from psychology and health science to economics and political ecology. During the three decades of study that produced this literature, public acceptance of wind energy "has gone from a marginal and little studied point of discussion to be at the forefront of broader debates in the social sciences" (Fournis & Fortin, 2016, p. 1). The study of wind energy acceptance has evolved considerably over this time. While early studies were exploratory and anecdotal, the statistical rigor of analysis, the application of diverse methods, and the development of complex theoretical frameworks have all improved over time.

Accordingly, the North American literature has coalesced around the need to understand two primary dependent variables or outcomes of interest related to wind energy acceptance: level of support/opposition, and attitudes. This paper distinguishes these two variables by using the terms "support" and "opposition" when discussing *proposed* or *hypothetical* wind facilities (pre-construction)

and "positive and negative attitudes" when discussing *existing* wind facilities (post-construction).⁴ Each of these variables has a range (e.g., from support to oppose or from positive to negative) that are typically expressed in five-point Likert scales. The literature has similarly coalesced around a number of overarching explanatory variables, which influence or explain changes in these dependent variables. The major explanatory variables in the North American wind acceptance literature are detailed in Section 5.

4. Limitations of Previous North American Wind Acceptance Research

This section describes a number of fallacies, biases, and limitations that have pervaded previous North American wind acceptance research. In some cases, these biases and limitations have prevented meaningful understandings and obstructed implementation of research findings and recommendations among developers and policymakers. Where appropriate, recommendations for future research are noted briefly, which are later summarized in sections 6 and 7.

From early surveys of residents near California wind farms (Bosley & Bosley, 1988) until today, opposition and negative attitudes toward wind energy have commonly been described by developers, politicians, and researchers as not-in-my-backyard (NIMBY) resistance (Petrova, 2013). The NIMBY concept posits that individuals favor wind energy generally but not in their local context—not in their "backyard." NIMBY also has been studied for decades in the context of nuclear energy and other hazardous facilities as well as social facilities such as prisons and mental health institutions (Wolsink, 2000).

However, many researchers now agree that the NIMBY framework is overly simplistic and unable to explain the complex motivations, concerns, and perceptions that can lead to opposition and negative attitudes (Devine-Wright, 2005; Petrova, 2013). Firestone et al. (2012b) stress that NIMBY resistance may be a *result* of opposition, rather than an explanation of it. Moreover, the term is generally used pejoratively (Kempton et al., 2005). A study in Texas concludes that NIMBY is "politically inappropriate and can often lead to misunderstanding, adding little value to the decision-making process" (Swofford & Slattery, 2010, p. 2516).

Social science researchers now generally agree that the language and concept of NIMBY as an explanation for wind energy opposition should be abandoned altogether⁵ (Devine-Wright, 2005; Petrova, 2013; Wolsink, 2006, 2012).

Another problem with the literature is positivist language toward wind energy, which some researchers

⁴ In some studies, residents near existing facilities are asked about their level of support for additional wind development in the area. The distinction between "attitudes" toward the existing facility and "support" for additional development still applies in these cases.

⁵ Based on ample evidence that the NIMBY explanation is problematic and unhelpful, this paper discards the term and instead focuses on examining other proposed explanations and correlates of wind energy acceptance.

have argued may reduce the quality and reliability of research and may prevent meaningful understandings of public acceptance (Aitken, 2010; Ellis et al., 2007). European researchers such as Aitken (2010), Ellis (2007), and Taebi (2016) have criticized researchers who portray wind energy opponents as "deviant" and seek to understand opposition merely to "overcome" it, but this criticism of positivist research has not been highlighted or examined to the same degree by North American researchers. It has been suggested that instead of focusing on the reasons for negativity toward wind energy, some researchers have sought methods to ensure approval (Taebi, 2016). According to Ellis et al. (2007, p. 536), this positivist lens has led to "poor explanatory findings, which in turn has resulted in ineffective policy." Songsore and Buzzelli (2014) stress that focusing only on mitigating resistance neglects important community concerns and may lead to negative consequences. Positivist language in the literature may include, for example:

- Statements that favor the needs of the wind industry over opponents, such as: "without public support in communities across the country, the industry's ability to build wind farms where it needs them may be hindered by nimbyism" (Klick & Smith, 2010, p. 1590)
- Statements that suggest opposition is a barrier that must be overcome and opponents are deviant, such as: "social barriers are blocking our way. That is to say, people are creating the problems, not technology" (Pasqualetti, 2011b, p. 202)
- Suggestions that the motivation of wind energy acceptance research is to help meet federal or state level renewable energy goals (e.g., Mulvaney et al., 2013)

Such positivist language has appeared regularly in the U.S. research to date (such as: Gipe, 1995; Klick & Smith, 2010; Olson-Hazboun et al., 2016; Pasqualetti, 2011b; Sovacool, 2009), but is less present in prominent Canadian research (such as: Baxter et al., 2013; Fast et al., 2016; Shaw et al., 2015; Walker et al., 2014b, 2014c), and rare in research rooted in humanistic geography (such as: Abbott, 2010; Phadke, 2013; Walker et al., 2014b), which tends to express more empathy toward affected communities. Researchers should cautiously avoid a positivist research lens.

Many North American wind energy acceptance studies have focused on hypothetical or proposed wind facilities, rather than existing facilities. Although asking individuals about their opinions toward hypothetical or proposed wind facilities can help answer certain research questions (e.g., about facilities, like offshore U.S. wind farms, that do not yet exist), these studies cannot reflect the unique feelings and experiences of residents living near existing turbines. Moreover, many studies around existing wind projects have not adequately examined the population living nearest to the turbines by oversampling and/or isolating the nearest residents for unique analyses (such as: Baxter et al., 2013; Bidwell, 2013; Fergen & Jacquet, 2016; Groth & Vogt, 2014; Olson-Hazboun et al., 2016; Pasqualetti & Butler, 1987; Petrova, 2014; Slattery et al., 2012; Thayer & Freeman, 1987). As such, numerous researchers have called for research focusing on residents closest to wind turbines (Hoen et al., 2011; Walker et al., 2014a). It is especially important that this group be represented in acceptance research, because it is most likely to be affected by wind facilities.

The vast majority of North American studies focus on only one or a few locations or wind facilities, so results cannot be generalized to the wider population living near wind turbines (e.g. Baxter et al., 2013;

Bidwell, 2013; Firestone & Kempton, 2007; Firestone et al., 2012b; Groothuis et al., 2008; Groth & Vogt, 2014; Olson-Hazboun et al., 2016; Pasqualetti & Butler, 1987; Petrova, 2014; Slattery et al., 2012; Thayer & Freeman, 1987). Some of these studies have used convenience samples rather than robust random samples, further limiting their external validity (e.g. Landry et al., 2012; Mulvaney et al., 2013b; Walker et al., 2014c). Fast and Mabee (2015, p. 27) suggest that this qualitative, case-study nature of wind acceptance research "does not translate well to conventional policy making." The dominance of these discrete case studies with poor comparability between them has recently led some European researchers to question whether wind acceptance research is "running out of steam" (Ellis & Ferraro, 2017).

There are considerable challenges and costs to developing and deploying research that is broadly representative across large regions like North America, making such studies out of reach for most researchers. Case studies may still add considerable insights and value in the North American context, but the value of these studies could be compounded through comparison. Future research might attempt to standardize some survey items and protocols in order to enable comparison of data across multiple case studies.

Nationwide surveys of wind acceptance in the United States and Canada ask only very broad questions, for example, about levels of support for wind energy generally. These broad surveys typically find high levels of support and positive attitudes (Ipsos, 2010; Leiserowitz et al., 2014; Vyas & Hurst, 2013), but they tell us little about *why* respondents feel the way they do. Some researchers have even called into question the validity of such general opinion polls (Aitken, 2010).

It would be useful to examine wind energy acceptance in concert with acceptance of other energy sources—like coal, nuclear, natural gas, and solar—but this has rarely been done. A notable exception is the work of Jeffery Jacquet, whose studies examine attitudes toward and perceived impacts of wind energy and natural gas developments, finding more polarized and negative attitudes and larger perceived impacts (both negative and positive) related to natural gas developments (Jacquet, 2012; Jacquet & Stedman, 2013).

The successful implementation of thirty years of research findings into developer and policymaker practice over the past has also been limited. As Zaunbrecher and Ziefle (2016, p. 312) state, "many factors that influence the social acceptance of wind power plants are already known. However, a conceptual framework for wind power plant planning that integrates these factors as well as the method of assessing them is still missing." Frameworks for wind project planning that increase community engagement and reduce conflict, such as those developed by Petrova (2016) and Jami & Walsh (2017), could continue to be developed and examined.

Despite these limitations, the North American wind acceptance literature has contributed significantly to the state of knowledge, and has evolved iteratively over the past 30 years, with improving rigor of research over time. The literature's major findings and contributions are summarized in the following section.

5. Overarching Aspects and Explanatory Variables in North American Literature

The North American literature reveals two primary dependent variables related to wind energy acceptance: level of support/opposition and attitudes among residents living near proposed or existing wind facilities. Among the explanatory variables that researchers correlate to those dependent variables, six overarching themes emerge: (1) socioeconomic aspects (including compensation); (2) sound annoyance and health risk perceptions; (3) visual/landscape aspects, annoyance, and place attachment; (4) environmental concerns and attitudes; (5) perceptions of planning process, fairness, and trust; and (6) distance from turbines (proximity hypothesis). Each theme is detailed below.

5.1 Socioeconomic Aspects

Individuals express a great deal of positive and negative concern over the socioeconomic aspects of wind facilities. Some studies find anticipated economic effects to be the variable most strongly influencing support or opposition to proposed wind developments as well as positive or negative attitudes toward existing sites (Bidwell, 2013; Brannstrom et al., 2011; Slattery et al., 2012; Songsore & Buzzelli, 2015).

5.1.1 **Positive economic aspects**

Positive economic aspects of wind energy development include rural economic development (Mulvaney et al., 2013b) including creation of jobs and other economic opportunities (Slattery et al., 2012), local tax revenue and/or lower tax rates for individuals (Slattery et al., 2012), increased tourism (Groth & Vogt, 2014), reduced electricity rates (Baxter et al., 2013) and landowner compensation (Jacquet, 2012). Landowner compensation, however, is not a universally positive socioeconomic impact for individuals living near turbines. It may create perceptions of "winners" and "losers" (Firestone et al., 2012b) and increase intra-community conflict (Baxter et al., 2013; Walker et al., 2014b). Compensation can even be seen as a form of bribery (Gipe, 1995). Having some form of compensation for nearby residents that are not hosting turbines on their land may lesson conflict and notions of winners and losers. For example, non-monetary, non-individual compensation such as the creation of dedicated wildlife habitats or support of community projects was supported by non-hosting community members in one study (Groth & Vogt, 2014). Other research suggests that non-hosting community members prefer public compensation over private compensation (García et al., 2016). Another form of compensation that has been examined is community investment in or ownership of wind facilities. Local ownership enables more equitable distribution of financial benefits as well as a higher degree of participation and influence in the development of a wind facility (Fast et al., 2016). This model has been shown to increase support in the European context, but little evidence exists in the North American context where community ownership remains rare (Bolinger, 2005; Ferguson-Martin & Hill, 2011; Sovacool & Ratan, 2012). In general, more research is needed to understand appropriate and acceptable compensation mechanisms for individuals and communities.

5.1.2 Negative economic aspects

Perceived negative socioeconomic impacts include reduced property values (Abbott, 2010; Firestone & Kempton, 2007; Hoen et al., 2015), decreased tourism (Landry et al., 2012; Lilley et al., 2010; Lutzeyer, 2013), increased traffic (Slattery et al., 2012), exacerbating economic inequality (Walker et al., 2014b, 2014c), impacts to fishing and other recreational opportunities (Firestone et al., 2009), and increased electricity rates (Baxter et al., 2013). Impacts on electricity rates are seen as a two-sided coin, with supporters citing reduced rates and opponents citing increased rates (Firestone et al., 2012a). Although nationwide and state-level studies in the United States have not found evidence of consistent, measurable, or significant reductions in home values near operating wind facilities (Hoen & Atkinson-Palombo, 2016; Hoen et al., 2015; Lang et al., 2014), the *perception* or belief of property value impacts may still affect acceptance of wind (Abbott, 2010; Walker et al., 2014a). Additionally, there is evidence that home-value effects might exist in the U.S. (Heintzelman & Tuttle, 2012) and Canadian (Fast et al., 2015) contexts, and there is growing evidence that effects exist in the European context (e.g. Dröes & Koster, 2016; Gibbons, 2015; Jensen et al., 2014). More research in this area could not only untangle conflicting results, but also increase understanding of how perceptions of property value impacts influence acceptance.

5.1.3 **Distributional justice**

The distribution of the costs and benefits of wind energy developments, broadly referred to as *distributional justice*, has been widely studied in the literature. Survey respondents consistently express concern that the energy and economic benefits produced from local wind facilities do not stay local and benefit local residents (Baxter et al., 2013; Groth & Vogt, 2014). Some studies have shown angst and opposition toward multinational corporate wind developers (Pasqualetti, 2011a; Petrova, 2013), and Firestone and Kempton (2007) demonstrate that support would increase for a proposed wind facility if it were being developed by the local government, rather than a private developer. The inability of local community members to invest or share ownership in wind energy developments has been cited as a factor in negative attitudes (Songsore & Buzzelli, 2015).

Unfair distribution of costs and benefits may lead to intra-community and/or rural-urban conflicts (Hirsh & Sovacool, 2013; Larson & Krannich, 2016; Pasqualetti, 2000; Phadke, 2013; Rule, 2014; Sovacool, 2009; Walker et al., 2014c) or injustices toward indigenous communities (Huesca-Pérez et al., 2016). Phadke (2013, p. 248) summarizes this conflict: "Rural communities at the forefront of new energy development are asking why they are disproportionately being asked to carry the weight of the new carbon economy while urban residents continue their conspicuous use of energy." Rural residents may also feel exploited by urban, multinational, corporate project developers seeking profits over public welfare (Petrova, 2013; Sovacool, 2009). Thus, some individuals who oppose or hold negative attitudes toward wind facilities may be fighting against a feeling of injustice as they find themselves on the front lines of development impacts while still on the margins of politics and economic opportunity. On the other hand, rural-urban conflicts may also propagate when the local, rural residents *support* the wind facility. Sovacool (2009, p. 4510) suggests that, in some cases, "rural [longstanding] residents want renewable power projects for their own use, as a vehicle for economic development, and resent what seems like meddling by urban [newly arrived] residents intent on preserving the countryside for its

scenic and recreational value."

Perceived socioeconomic impacts are at the forefront of concerns for many individuals living near existing and proposed wind facilities, but those perceived impacts and the ways they influence acceptance are complex. More research is needed to understand inter- and intra-community conflicts, the effects of and community responses toward compensation mechanisms, and the relationships between perceived economic impacts and perceived fairness of planning processes and outcomes.

5.2 Sound Annoyance and Health Risk Perceptions

5.2.1 Annoyance from wind turbine sound

Some studies have correlated turbine sound with *annoyance*, which may be associated with sleep disturbance, negative emotions, or other health-related effects (Knopper & Ollson, 2011; Knopper et al., 2014; Michaud et al., 2016a). The annoyance experienced by people living near utility-scale wind facilities is correlated to more negative attitudes (Fast et al., 2016; Firestone et al., 2015). This annoyance, however, may be more strongly correlated to other characteristics rather than wind turbine sounds (McCunney et al., 2014). In Europe, Pedersen & Waye (2004), showed that residents' level of annoyance with wind turbine sound is strongly influenced by their level of annoyance with the visual impact of turbines (discussed in section 5.3.1), yielding higher annoyance with turbine sound compared with dose-response curves from other, non-turbine sound emissions, such as transportation noise. This result deserves further study in the North American context.

Some research has demonstrated that annoyance and complaints decline with increased distance from turbines (Kaliski & Neeraj, 2013; Nissenbaum et al., 2012), but there is no general consensus about the setback distance required to minimize or mitigate annoyance (Nissenbaum et al., 2012) as distance is just one component of how sound from turbines propagates to nearby residents. Accordingly, researchers (and stakeholders in general) often rely on a sound-specific threshold to reduce annoyance and stress impacts and concerns from local residents, which is commonly 40-45 dBA⁶ (Knopper & Ollson, 2011; Knopper et al., 2014; Paller, 2014; Phadke, 2013). The World Health Organization recommended a maximum of 45 dBA outside of homes during night hours (World Health Organization, 1999), but that recommendation was revised for the European Union in 2009 to 40 dBA (World Health Organization, 2009). This sound level has been compared to the sound level of a quiet office, library, a computer, or a refrigerator in a nearby room⁷. In a recent comprehensive study of measured wind turbine sound levels and reported health effects, turbine noise reached a maximum of 46 dBA and a mean of 35.6 dBA for 1,238 residents living between 0.25 – 11.22 kilometers from operational wind turbines in Canada (Michaud et al., 2016b).

Although sound levels comparable to a quiet office or library may not be annoying to most people, studies have suggested that wind turbine noise is considered annoying at much lower sound levels than those causing annoyance from other sources (Janssen et al., 2011). Some recent research has attributed

⁶ dBA = A-weighted decibels, a measure of loudness as perceived by the human ear. Measurements are typically average nighttime levels outside homes, and do not include ambient noise.

⁷ For example dBA comparisons, see e.g., http://www.rlcraigco.com/pdf/dba-comparison.pdf

this to the amplitude modulation (i.e. swishing or thumping) of turbine sounds (Yoon et al., 2016), however this has not been rigorously examined in the North American context.

5.2.2 Health effects of wind turbine sound

Recent epidemiological research concludes that wind turbine sound and infrasound⁸ are not directly related to adverse human health effects (Knopper & Ollson, 2011; Knopper et al., 2014; Michaud et al., 2016a) or sleep quality (Michaud et al., 2016b). Some research attributes wind-related health symptoms to the "nocebo" hypothesis, in which the *expectation* of negative health effects influences symptoms experienced (Knopper et al., 2014).

Nonetheless, the *perception* of health risk clearly reduces support for wind facilities (Baxter et al., 2013; Magari et al., 2014), and some wind opponents may feel that potential health risks are not adequately addressed (Songsore & Buzzelli, 2014). Walker et al. (2014c, p. 1) suggest a move beyond debating "whether or not 'annoyance' represents a 'health impact' and instead focus[ing] on ways to minimize ... feelings of threat and stress at the community level." Similarly, Fast et al. (2016, p. 3) conclude that "rather than dismissing health claims as groundless or inconsequential, policy-makers should take a precautionary approach so as to more thoroughly address the factors that contribute to frustration on the part of host communities." If community concerns and expectations regarding sound and health impacts are not adequately addressed, a portion of the population may remain annoyed even after noise limits are enforced (Knopper et al., 2014).

Although there is a demonstrated correlation between wind facility sound, annoyance, and negative attitudes, more research is needed to understand these relationships. Topics that must be explored include measured (or modeled) sound and reported annoyance levels; types of sounds that are particularly annoying; the relationships among sound, annoyance, and respondents' distance from turbines; and the influence of other variables such as visual annoyance, place attachment, procedural fairness, and respondent characteristics.

5.3 Visual/Landscape Aspects, Annoyance, and Place Attachment

Visual impacts and landscape change are some of the most frequently cited correlates to reduced support of proposed wind projects and negative attitudes toward existing wind facilities. In general, visual and landscape concerns relate to a desire of communities to protect local landscape quality and identity in the face of change (Phadke, 2010).

5.3.1 Aesthetics and Annoyance

Numerous studies have indicated that the diminution of scenic beauty due to existing or proposed wind facilities may contribute to annoyance and is often linked to negative attitudes or reduced support (Bosley & Bosley, 1988; Bush & Hoagland, 2016; Gipe, 1993, 1995; Jacquet & Stedman, 2013; Pasqualetti & Butler, 1987; Pasqualetti et al., 2002; Phadke, 2010; Rule, 2014). Visual annoyance may

⁸ Infrasound is sound at frequencies lower than 20 Hz, which may be emitted by wind turbines as well as a number of other sources.

also result from "shadow flicker" created near turbines as sunlight passes through the blades of wind turbine in motion (Rule, 2014). Knopper & Ollson (2011) conclude that annoyance and self-reported health effects are more strongly related to visual impacts than to sound from wind facilities. Some researchers have suggested guidelines or best-practices for minimizing visual impacts of wind developments (Apostol et al., 2016; Pasqualetti et al., 2002); visual impact guidelines have also been suggested by U.S. (National Academy of Sciences (NAS), 2007; Sullivan et al., 2012) and Canadian (BC Ministry of Forests, 2016) government organizations.

New wind development in North America now routinely includes some form of visual impact assessment, typically in the form of computer-generated visual simulations of what a wind facility may look like from various vantage points (Apostol et al., 2016; Phadke, 2010). Phadke (2010, p. 17) argued that the visualizations created by wind energy developers and project opponents alike are an "immature policy craft", are inherently political, and encoded with cultural values. As such, they may further polarize opinions rather than providing useful information to community stakeholders. However, Apostol et al. (2016) suggest useful guidance and techniques to improve such visualizations and enhance their usefulness.

Negative attitudes stemming from the visual impacts of wind turbines may not occur simply because people dislike how turbines look; people also have become accustomed to an electricity system that is essentially "invisible" to consumers owing to centralized infrastructure typically sited far from population centers (Pasqualetti, 2000; Sovacool, 2009). As Sovacool (2009, p. 4509) states, "the physical 'removal' of power stations from most cities and neighborhoods also 'removes' them from the minds of most Americans, and contributes to public apathy and misunderstanding." Until recently, "electric generators were usually built in obscure locations, perceptible only to a few people. But wind turbines, by their very nature, require a highly dispersed and visible distribution, often in attractive and unspoiled areas" (Hirsh & Sovacool, 2013, p. 724).

On the other hand, visual impacts are not universally negative; there is some evidence for positive visual and symbolic perceptions from wind facilities (e.g., Brannstrom et al., 2011; Firestone et al., 2015; Mulvaney et al., 2013a; Phadke, 2010).

The rotational motion of wind turbines has been another topic of study. Early studies found that perceptions that turbines were "unreliable" or often not operating were correlated to negative attitudes and concerns about tax fraud (Gipe, 1993; Pasqualetti & Butler, 1987; Thayer & Freeman, 1987). More recently, Fergen and Jacquet (2016) found that respondents believed nearby turbines were more beautiful when the turbines were in motion, which they attribute to notions of economic productivity of turbines in motion compared to lost economic opportunity of motionless turbines.

5.3.2 Visual impacts and economics

Some researchers have highlighted visual impacts from wind turbines in choice experiments or other economic modeling techniques. For example, some property value impact studies use distance and views of the turbines as correlates (e.g. Hoen et al., 2011; Jensen et al., 2014). Some residents may be willing to pay to minimize the perceived negative visual impacts of proposed wind facilities (Boatwright,

2013; Groothuis et al., 2008; Krueger et al., 2011; Pasqualetti, 2001). Visual impacts have even been framed as a property-rights infringement in some cases (Abbott, 2010).

5.3.3 Place Attachment

Threats to *place attachment*—an emotional bond between individuals and the familiar locations they inhabit—are highlighted as a correlate in European literature (Devine-Wright, 2009; Devine-Wright & Howes, 2010). Under the place attachment theory, landscape impacts extend beyond aesthetics into the identities, connections, and meanings that individuals form with a particular location (Devine-Wright, 2009). Some North American studies also emphasize the role of place attachment in influencing wind energy acceptance (Bidwell, 2013; Fast & Mabee, 2015; Firestone et al., 2009), but Jacquet and Stedman (2013) found that place meanings seemed to have little or no association with acceptance of wind development in their Pennsylvania study. Future research in North America should continue to examine the influence of place attachment on acceptance.

Visual impacts are a widely studied, well-documented correlate to wind energy acceptance. Requiring further study, however, are the extent to which visual impacts influence other explanatory variables and vice versa (e.g., sound); the mechanisms behind *positive* visual perceptions; the effects of distance and physical geography on visual annoyance; the effect of different degrees of visual impacts (such as seeing the full sweep of turbine blades from the home vs. only a portion of the turbine); the frequency individuals see the turbines; and the role of planning process fairness and/or participation in diminishing or mitigating visual annoyance.

5.4 Environmental Concerns and Attitudes

Numerous researchers have noted that, in wind power siting debates, both supporters and opponents base their arguments on environmental concerns. These so-called "green vs. green" debates typically revolve around local environmental harms (e.g., wildlife, landscape, and noise impacts) versus regional or global benefits (e.g., climate change mitigation and air pollution reduction) (Firestone et al., 2009; Warren et al., 2005). As such, some studies have found pro-environmental beliefs and values to correlate positively with support for wind energy (Larson & Krannich, 2016; Mulvaney et al., 2013b), while others have found the opposite (Fergen & Jacquet, 2016). Research has also shown that environmental beliefs may correlate to support for wind energy broadly, but that support may not exist when the same individuals are asked about local wind energy development (Larson & Krannich, 2016). In some cases, individuals with stronger environmental attitudes may prioritize the conservation of "natural" landscape over the benefits of renewable energy (Fergen & Jacquet, 2016). Abbott (2010, p. 971) concludes that these multiple conservation priorities perpetuate environmental conflicts in local contexts.

5.4.1 Wildlife concerns

Since the earliest wind facilities in North America began operation, the potential threats of wind energy to wildlife, particularly birds and bats, have been of significant concern. A 2013 study estimated that between 140,000 and 328,000 birds are killed annually by collisions with wind turbine towers in the contiguous United States (Loss et al., 2013). This is a notable impact that may reduce support for wind

energy in some individuals. However, Sovacool (2013) estimates that wind energy kills approximately 13 times fewer birds than fossil fueled power plants per kilowatt-hour of electricity generated. Similarly, a recent Canadian study found that avian mortality due to wind turbines was relatively small compared to other sources of human-influenced avian mortality (Zimmerling et al., 2013). These factors may induce some individuals to prefer the avian impacts of wind energy in comparison to alternatives (thus increasing support). Perhaps due to these conflicting considerations, previous research does not demonstrate a clear influence of wildlife impacts on onshore wind energy acceptance. Summary statistics have shown between 18-24% of local residents consider onshore wind turbines dangerous to wildlife (Mulvaney et al., 2013a; Slattery et al., 2012; Thayer & Freeman, 1987), and few studies have found this concern to statistically affect acceptance (e.g., Larson & Krannich, 2016).

Experience from the Cape Wind project suggests that wildlife concerns may significantly influence acceptance of *offshore* wind energy. Firestone et al. (2012a), for example, found that 48% of respondents believed Cape Wind would cause harm to bird life, and 44% thought it would harm marine life (those percentages decreased slightly in a repeat survey in 2009). Firestone and Kempton (2007) reported that if Cape Wind were found to harm marine or bird life, the majority of respondents would be less likely to support the project. In their book on Cape Wind, Williams and Whitcomb (2007) also emphasize the role of perceived impacts to birds, fish, and whales in shaping public opinion.

5.4.2 **Climate change concerns**

Wind energy's potential to mitigate climate change is a benefit often cited by supporters (Petrova, 2013), but concern for climate change alone does not fully explain support for wind. Accordingly, efforts to encourage support by emphasizing climate benefits may be met with indifference (Bidwell, 2015; Firestone et al., 2009; Petrova, 2016; Swofford & Slattery, 2010). Olson-Hazboun et al. (2016, p. 168) further suggest that emphasizing environmental and climate benefits may actually *increase* opposition in some contexts owing to the politically polarizing nature of such topics; they conclude that "the framing of renewable energy as an environmental issue could have unintended and adverse effects in certain social and political contexts." Other research has found that even people who are unconcerned about the environment or the use of fossil fuels may strongly support wind energy development based on potential economic opportunities (Slattery et al., 2012).

Overall, a number of researchers have found that support for wind energy is less correlated to environmental beliefs than to other factors such as economic and landscape impacts (Olson-Hazboun et al., 2016). Nonetheless environmental concerns clearly play a role in attitude formation for many individuals living near turbines, and more research could add nuance to the perceived environmental trade-offs of wind energy. Future work should continue to examine the role of environmental motivations for wind energy attitudes in different socio-political contexts.

5.5 Perceptions of Planning Process, Fairness, and Trust

The processes around wind project planning and development can significantly affect public acceptance (Firestone et al., 2012b), and a lack of opportunity for local residents to engage meaningfully in the planning process may reduce support or increase local conflict (Bohn & Lant, 2009; Huesca-Pérez et al.,

2016; Phadke, 2011). In some wind development models, local citizens have been entirely removed from the planning and design of wind developments (Phadke, 2013). This may lead to feelings of injustice among local residents, who perceive that "government and corporate decision-making ... takes place in faraway boardrooms disconnected from the people and landscapes that will be directly affected" (Phadke, 2013, p. 247). This perception of injustice may be especially severe among already disadvantaged communities (Huesca-Pérez et al., 2016).

A more participatory, collaborative planning process may reduce conflict and promote positive community outcomes (Groth & Vogt, 2014; Jami & Walsh, 2017; Songsore & Buzzelli, 2015; Walker et al., 2014c). Some scholars have suggested moving away from a "decide-announce-defend" model of wind facility planning toward a more collaborative, "consult-consider-modify-proceed" process (Phadke, 2013, p. 250). Indeed, Jacquet (2015) found that landowners participating and negotiating in the development process were better informed and more supportive than non-participating landowners. Some researchers, however, have demonstrated significant barriers to genuinely participatory, consensus-based planning processes, which may prevent widespread implementation of such strategies (Jami & Walsh, 2014). The numerous calls from researchers to increase public participation rarely include detail on how to implement participatory methods or measure their success (Bidwell, 2016b). In response to this need, a recent paper developed a recommended framework for meaningful community engagement and outlined a number of suggested strategies to increase public participation and reach consensus, including: early involvement of the community, being available, proactive, and present within the community, building relationships and trust, and offering financial investment in the project to the local community (Jami & Walsh, 2017). Nonetheless, questions still remain about how and when to engage the public and encourage participation in the planning process. Despite the challenges, it is seen as counterproductive to exclude participation based on the assumption that local community members lack the necessary information to make informed decisions (Petrova, 2016).

A planning process perceived as "fair" can lead to greater toleration of the outcome, even if it does not fully satisfy all stakeholders (Firestone et al., 2012b), whereas processes perceived as "unfair" can result in conflict, damaged relationships, and divided communities (Fast et al., 2016). However, greater toleration is not necessarily synonymous with support or "successful coexistence" (Songsore & Buzzelli, 2014). In other words, individuals may tolerate an outcome they perceive as fair, even if they did not get exactly what they wanted. Owing to this distinction, some researchers have begun to study support and toleration as separate dependent variables (Petrova, 2013).

The perceived fairness of the planning process is linked to *trust* between the local community and the project developers, and some researchers consider this trust important for project support (Dwyer, 2016; Fast & Mabee, 2015). Shaw et al. (2015) conclude that public trust has been eroded by governments that have not effectively engaged communities in fair decision-making processes. Aitken (2010, p. 1840) stresses that a pro-wind bias among developers and policymakers can undermine trust among stakeholders: "In order for this trust to be meaningful it cannot be conceived as a means to a particular end—i.e. less opposition and more wind farms."

Although numerous studies have shown correlations among perceived inclusiveness, fairness, trust, toleration, and support, more research is needed to understand when and how actually to *implement* more collaborative, democratic planning processes. Future research should also further examine the relationships between perceived procedural justice and socioeconomic impacts of wind development.

5.6 Distance from Turbines (Proximity Hypothesis)

Since the earliest studies on this topic, researchers have consistently examined the hypothesis that those living closest to turbines will have the most negative attitudes about the local wind facility. These studies, however, have produced no clear consensus. Some studies confirm the notion that positive attitudes increase with distance from the nearest turbine (Swofford & Slattery, 2010; Thayer & Freeman, 1987), while others show the exact opposite: that those living nearest turbines have more positive attitudes and are less concerned about potential negative impacts of the turbines (Baxter et al., 2013; Groth & Vogt, 2014; Warren et al., 2005). However, it is unclear whether such results adequately account for confounding variables, such as landowner compensation and other economic benefits accruing to nearby residents.

Olson-Hazboun et al. (2016) studied a different, related variable in addition to distance from the nearest turbine: how frequently respondents see (or expect to see) wind turbines, which they call "visual accessibility." These authors found no significant relationship between distance and attitudes toward the wind facility, but they did find that residents who see wind turbines more frequently were less likely to express positive attitudes toward the facility. Other studies have proposed that impacts from wind facilities may be cumulative, increasing with the size of turbines, the number of turbines visible, and the clustering of turbines (Petrova, 2013; Walker et al., 2014c). However, some European studies have not found a significant correlation between the number of turbines and negative attitudes (Krohn & Damborg, 1999; Pohl et al., 2012). Questions around cumulative impacts and visual accessibility merit additional study in the North American context.

Some researchers have hypothesized that, over time, individuals will "self-sort," as those with more positive attitudes move closer to turbines and those with more negative attitudes move away (Hoen et al., 2015; Tiebout, 1956). This effect, however, has not been rigorously examined.

Although many researchers have found a correlation between distance from an individual to the nearest turbine and attitudes toward wind energy, the direction and strength of that correlation remains in question, as does the extent to which regional, demographic, or other factors may influence this correlation. More research is also needed to understand relationships between distance and other correlates to acceptance, such as sound and visual impacts.

5.7 Other Proposed Correlates of Acceptance

Aside from the six dominant themes discussed in the preceding sections, the literature identifies a number of other potential correlates of wind energy acceptance. Some researchers have proposed that a lack of knowledge about energy generally or wind energy specifically may explain opposition or negative attitudes (Bosley & Bosley, 1988; Bush & Hoagland, 2016; Klick & Smith, 2010; Sovacool,

2009), but this "information deficit" explanation has been largely discredited. Opponents and those with negative attitudes are not ignorant of wind energy facts (Fast, 2015), and high levels of knowledge about energy do not necessarily correlate with support or positive attitudes (Baxter et al., 2013). On the other hand, Bidwell (2016a) did find a relationship between an informational intervention and increased support, and suggests that the information deficit model should not be dismissed.

Demographic data are also routinely collected and examined as possible correlates to wind energy acceptance in survey research. However, throughout the literature, demographic variables such as gender, income, and education level do little to explain variation in wind energy support or attitudes. Where reported, the effects of demographic variables on acceptance are typically not statistically significant (e.g., Firestone et al., 2015; Jacquet & Stedman, 2013; Mulvaney et al., 2013b).

Other proposed explanations for acceptance include concerns about dependence on foreign energy sources (Firestone et al., 2009), personal and moral values (Bidwell, 2013, 2015), attitudes toward local or federal government policy (Fast & Mabee, 2015; Petrova, 2014; Songsore & Buzzelli, 2014), and the degree to which expectations about a development are met (Fergen & Jacquet, 2016).

Many of these additional variables may relate to and be influenced by the six major themes previously discussed. In addition to clarifying the influence of the six major themes, future research should explore the influence of these other variables.

6. Lessons learned in 30 years of wind acceptance studies

Over the past three decades, scholars have substantially advanced the state of knowledge about wind energy acceptance in North America. Each new study has added evidence to existing hypotheses, proposed new hypotheses, presented new methods for engaging stakeholders in research, used new methods for data analysis, and/or incorporated new geographic areas under the umbrella of research. The studies have collected data spanning the period before any local development to well after the wind facilities began operating. Drawing from the sections above, the lessons learned over the past 30 years are summarized here:

Overall, support is high, and attitudes are largely positive. The North American literature consistently finds favorable views of wind energy; 70%–90% of North Americans approve of wind energy generally, and support has been high for specific existing and proposed wind facilities as well.

Researchers should cautiously avoid a positivist research lens. Viewing opposition merely as something to be *overcome* reduces the quality of research and prevents meaningful understandings and implementation of best practices. The motivation of wind energy acceptance research should not be exclusively to ensure approval of wind energy developments.

NIMBY is invalid. The NIMBY explanation has been widely discredited as simplistic, pejorative, politically inappropriate, and unhelpful as a framework to explain public attitudes toward wind facilities both before and after they are built. Nonetheless, use of the term persists among the wind industry,

policymakers, even researchers.

Incorporating research into practice has lagged. Research over the past 30 years has produced many important insights, but these lessons have been slow to transition into practice among developers and policymakers.

Perceptions of turbine performance and reliability matter. Early studies revealed widespread concerns about turbine performance and reliability. More recently, studies have found a preference for turbines in motion compared to static turbines.

Demographics do not explain much. Throughout the literature, demographic variables such as gender, income, and education level do little to explain variation in wind energy attitudes; some studies have shown contradictory evidence related to these variables.

Socioeconomic impacts are very important. Local stakeholders are concerned with various socioeconomic impacts, and some researchers have found socioeconomic concerns to be paramount among local residents. In general, those living near wind facilities want benefits that stay in the local community, and they feel a sense of injustice about bearing the burden of costs when consumption of and profits related to the power are enjoyed elsewhere.

Sound and visual impacts are strongly tied to annoyance and opposition. Annoyance and opposition related to actual or expected sound and visual impacts are well documented throughout the literature. In some cases, annoyance and other impacts are ignored, downplayed, or assumed to be absent or inconsequential by developers and policymakers, which may exacerbate conflict and distrust among community members.

Environmental concerns matter, though perhaps less than other factors. Environmental concerns correlate with wind energy acceptance, but the strength of that correlation may be lower than that of other factors like socioeconomic impacts. Also, the direction of the correlation remains unclear.

Process fairness, participation, and trust can influence acceptance. A planning process that is perceived as "fair" can lead to greater toleration of the outcome, even if it does not fully satisfy all stakeholders. More participatory processes may increase trust and support, and ongoing post-construction community stewardship should be maintained.

Distance from turbines affects other variables, but alone its influence is unclear. The "proximity hypothesis" has yielded confounding findings in the literature. What is known is that an individual's distance from existing turbines affects a number of other correlates, including visual, sound, and socioeconomic impacts.

Other variables also affect acceptance, and the understanding of these is evolving. Researchers have proposed a wide range of other variables potentially correlated to wind energy acceptance, many of which deserve further study. Over time the addition of more possible correlates adds to the depth of understanding in this field.

7. Knowledge gaps after 30 years: Areas for future research

A number of questions remain unresolved after three decades of research in this field. The following are specific areas for future research.

A widespread, representative, random survey should be conducted in North America. Previous studies on acceptance of wind energy in North America have focused on only one or a few wind facilities. A more geographically representative survey examining the variables discussed in this paper would greatly advance the state of knowledge in this field.

Individuals living closest to turbines should be oversampled. The detailed experiences and attitudes of those living closest to turbines (i.e., within half a mile) have not been well captured. Future research should oversample this group and analyze their responses as a group that is distinct from those living further away.

Comparability of case studies should be enhanced. Discrete case studies should utilize some standardized survey items and protocols to facilitate comparison with data collected at different sites by other researchers.

Causation should be identified, where possible. Many studies have used regression techniques to tease out competing correlates of acceptance, but in many cases the *direction* of influence (i.e., causality) of these correlates is not understood.

Wind energy acceptance should be compared with acceptance of other energy sources. Only a few North American studies have examined wind acceptance in concert with acceptance of other energy sources (e.g. Jacquet, 2012). Future research should attempt to do so to provide a mechanism for comparison.

Change in acceptance over time should be studied. Longitudinal studies of acceptance over time (i.e., before, during, and after wind project construction) have revealed interesting changes in the European literature (e.g. Wolsink, 2007), but they have rarely been conducted in North America—a notable exception being Firestone et al. (2012a), but that study only covered the development and planning periods, leaving construction and post-construction periods in question. Similarly, the attitudes among respondents who moved into an area *after* the wind facility began operation have not been studied in depth.

Annoyance survey data should be combined with measured or modeled sound-level data. A number of surveys have asked respondents about their level of annoyance and perceived health impacts from wind turbine sound, but very few (e.g. Magari et al., 2014) have correlated those data to measured or modeled sound-level data.

Varying development-process models and experiences should be studied. Participation in, perceptions of, and resident preferences for the wind facility development process are not well understood.

Compensation mechanisms should be compared more rigorously. More research is needed to understand appropriate and acceptable compensation mechanisms for individuals and communities, such as local ownership and investment opportunities, community compensation schemes, and non-monetary community benefits.

Resident perceptions of property-value impacts should be studied in greater depth. Although recent large-scale research has not found a significant property-value impact on homes near wind facilities, those impacts may exist in some cases, and the *perception* of value impacts among local residents could exist, but is not well understood.

Implementation of research recommendations should be studied. Research is needed to understand the effects and implementation of strategies proposed in the literature during the development and policy-making process. Researchers should continue to develop frameworks for wind project planning that increases community engagement and reduces conflict and injustice.

8. Conclusion

The efforts of wind energy acceptance researchers over the past 30 years have yielded many important lessons and insights, but much work remains to be done in this space—particularly in the North American context. Thirty years from today, wind energy could conceivably generate over 30% of North America's electricity, representing a significant increase over the current installed capacity. More turbines will result in more nearby "neighbors," increasing the population that experiences the direct impacts of wind energy. Opposition and negative attitudes will, undoubtedly, still exist. However, with continued research and a commitment to implement research findings into developer and policymaker practice, conflict and perceived injustices might be significantly reduced.

References

- Abbott, J. A. (2010). The localized and scaled discourse of conservation for wind power in Kittitas County, Washington. *Society and Natural Resources*, *23*(10), 969-985.
- Aitken, M. (2010). Why we still don't understand the social aspects of wind power: A critique of key assumptions within the literature. *Energy Policy*, *38*(4), 1834-1841. doi: <u>http://dx.doi.org/10.1016/j.enpol.2009.11.060</u>
- Apostol, D., Palmer, J., Pasqualetti, M., Smardon, R., & Sullivan, R. (2016). *The Renewable Energy Landscape: Preserving Scenic Values in our Sustainable Future*: Routledge.
- AWEA. (2003). Global Wind Energy Market Report. Retrieved 4/24/17, from https://web.archive.org/web/20100918230645/<u>http://www.awea.org/pubs/documents/global</u> <u>market2003.pdf</u>
- Baxter, J., Morzaria, R., & Hirsch, R. (2013). A Case-Control Study of Support/Opposition to Wind Turbines: The Roles of Health Risk Perception, Economic Benefits, and Community Conflict. *Energy Policy*, *61*, 931-943.
- BC Ministry of Forests. (2016). A Visual Effects Assessment Guidebook for Wind Energy Developments in British Columbia: British Columbia Ministry of Forests, Lands, and Natural Resource Operations.

- Bidwell, D. (2013). The role of values in public beliefs and attitudes towards commercial wind energy. *Energy Policy, 58*, 189-199.
- Bidwell, D. (2015). *Public acceptance of offshore wind energy: Relationships among general and specific attitudes.* Paper presented at the OCEANS'15 MTS/IEEE Washington.
- Bidwell, D. (2016a). The Effects of Information on Public Attitudes Toward Renewable Energy. *Environment and Behavior, 48*(6), 743-768. doi: 10.1177/0013916514554696
- Bidwell, D. (2016b). Thinking through participation in renewable energy decisions. *Nature Energy*, 1(5), 16051. doi: 10.1038/nenergy.2016.51
- Boatwright, J. A. (2013). *Siting Community Wind Farms: An Investigation of NIMBY.* (Master's Degree in Agricultural and Applied Economics), Virginia Polytechnic Institute and State University, Blacksburg, VA.
- Bohn, C., & Lant, C. (2009). Welcoming the Wind? Determinants of Wind Power Development Among U.S. States. *The Professional Geographer*, *61*(1), 87-100. doi: 10.1080/00330120802580271
- Bolinger, M. A. (2005). Making European-Style Community Wind Power Development Work in the US. *Renewable and Sustainable Energy Reviews, 9*(6), 556-575.
- Bosley, P., & Bosley, K. (1988). Public acceptability of California's wind energy developments: three studies. *Wind Engineering*, *12*(5), 311-318.
- Brannstrom, C., Jepson, W., & Persons, N. (2011). Social Perspectives on Wind-Power Development in West Texas. Annals of the Association of American Geographers, 101(4), 839-851. doi: 10.1080/00045608.2011.568871
- Bush, D., & Hoagland, P. (2016). Public opinion and the environmental, economic and aesthetic impacts of offshore wind. *Ocean & Coastal Management, 120,* 70-79. doi: 10.1016/j.ocecoaman.2015.11.018
- CanWEA. (2016). Canada's Wind Energy Story. Retrieved January 27, 2017, from <u>http://canwea.ca/wp-content/uploads/2017/01/canadas-wind-energy-story-2016-e-v3.pdf</u>
- Corscadden, K., Wile, A., & Yiridoe, E. (2012). Social license and consultation criteria for community wind projects. *Renewable Energy*, 44, 392-397. doi: 10.1016/j.renene.2012.02.009
- Devine-Wright, P. (2009). Rethinking NIMBYism: The role of place attachment and place identity in explaining place-protective action. *Journal of Community & Applied Social Psychology, 19*(6), 426-441.
- Devine-Wright, P., & Howes, Y. (2010). Disruption to place attachment and the protection of restorative environments: A wind energy case study. *Journal of environmental psychology, 30*(3), 271-280. doi: 10.1016/j.jenvp.2010.01.008
- Devine-Wright, P. (2005). Beyond NIMBYism: towards an integrated framework for understanding public perceptions of wind energy. *Wind energy*, *8*(2), 125-139.
- Dröes, M. I., & Koster, H. R. A. (2016). Renewable energy and negative externalities: The effect of wind turbines on house prices. *Journal of Urban Economics*, 96, 121-141. doi: 10.1016/j.jue.2016.09.001
- Dwyer, J. (2016). *Perceptions of the Block Island Wind Farm Process: Perspectives from Those Involved.* (Master of Arts), University of Rhode Island.
- Ellis, G., Barry, J., & Robinson, C. (2007). Many ways to say 'no', different ways to say 'yes': applying Qmethodology to understand public acceptance of wind farm proposals. *Journal of environmental planning and management*, *50*(4), 517-551.
- Ellis, G., & Ferraro, G. (2016). The social acceptance of wind energy: Where we stand and the path ahead. In E. Commission (Ed.).
- Ellis, G., & Ferraro, G. (2017). *Has research on the social acceptance of wind energy run out of steam?* Paper presented at the 1st International Conference on Energy Research and Social Science, Sitges, Spain.

- Fast, S. (2015). Social Science Explanations for Host Community Responses to Wind Energy. In Leal (Ed.), Handbook of Renewable Energy (pp. 1-15). Berlin: Springer.
- Fast, S., & Mabee, W. (2015). Place-making and trust-building: The influence of policy on host community responses to wind farms. *Energy Policy*, *81*, 27-37.
- Fast, S., Mabee, W., Baxter, J., Christidis, T., Driver, L., Hill, S., McMurtry, J. J., & Tomkow, M. (2016). Lessons learned from Ontario wind energy disputes. *Nature Energy*, 1(2), 15028. doi: 10.1038/nenergy.2015.28
- Fast, S., Mabee, W., & Blair, J. (2015). The changing cultural and economic values of wind energy landscapes. *The Canadian Geographer / Le Geographe canadien, 59*(2), 181-193.
- Fergen, J., & Jacquet, J. (2016). Beauty in motion: Expectations, attitudes, and values of wind energy development in the rural U.S. *Energy Research & Social Science*, 11, 133-141. doi: 10.1016/j.erss.2015.09.003
- Ferguson-Martin, C. J., & Hill, S. D. (2011). Accounting for variation in wind deployment between Canadian provinces. *Energy Policy*, *39*(3), 1647-1658. doi: 10.1016/j.enpol.2010.12.040
- Firestone, J., Bates, A., & Knapp, L. A. (2015). See me, Feel me, Touch me, Heal me: Wind turbines, culture, landscapes, and sound impressions. *Land Use Policy*, 46, 241-249. doi: 10.1016/j.landusepol.2015.02.015
- Firestone, J., & Kempton, W. (2007). Public opinion about large offshore wind power: Underlying factors. *Energy Policy*, *35*(3), 1584-1598. doi: 10.1016/j.enpol.2006.04.010
- Firestone, J., Kempton, W., & Krueger, A. (2009). Public Acceptance of Offshore Wind Power Projects in the USA. *Wind energy*, *12*(2), 183-202.
- Firestone, J., Kempton, W., Lilley, M. B., & Samoteskul, K. (2012a). Public acceptance of offshore wind power across regions and through time. *Journal of environmental planning and management*, *55*(10), 1369-1386.
- Firestone, J., Kempton, W., Lilley, M. B., & Samoteskul, K. (2012b). Public acceptance of offshore wind power: does perceived fairness of process matter? *Journal of environmental planning and management*, *55*(10), 1387-1402.
- Fournis, Y., & Fortin, M.-J. (2016). From social 'acceptance' to social 'acceptability' of wind energy projects: towards a territorial perspective. *Journal of environmental planning and management*, 60(1), 1-21. doi: 10.1080/09640568.2015.1133406
- García, J. H., Cherry, T. L., Kallbekken, S., & Torvanger, A. (2016). Willingness to accept local wind energy development: Does the compensation mechanism matter? *Energy Policy, 99*, 165-173. doi: 10.1016/j.enpol.2016.09.046
- Gibbons, S. (2015). Gone with the wind: Valuing the visual impacts of wind turbines through house prices. *Journal of Environmental Economics and Management, 72*, 177-196. doi: 10.1016/j.jeem.2015.04.006
- Gipe, P. (1993). The wind industry's experience with aesthetic criticism. *Leonardo*, *26*(3), 243-248.
- Gipe, P. (1995). Wind Energy Comes of Age. New York, NY: Wiley Press.
- Global Wind Energy Council (GWEC). (2017, 2/10/17). Global Wind Statistics 2016. from http://www.gwec.net/wp-content/uploads/vip/GWEC_PRstats2016_EN_WEB.pdf
- Groothuis, P. A., Groothuis, J. D., & Whitehead, J. C. (2008). Green vs. green: Measuring the compensation required to site electrical generation windmills in a viewshed. *Energy Policy*, *36*(4), 1545-1550.
- Groth, T. M., & Vogt, C. (2014). Residents' perceptions of wind turbines: An analysis of two townships in Michigan. *Energy Policy*, *65*, 251-260. doi: 10.1016/j.enpol.2013.10.055
- Heintzelman, M. D., & Tuttle, C. (2012). Values in the Wind: A Hedonic Analysis of Wind Power Facilities. *Land Economics, August* (88), 571-588.

- Hirsh, R. F., & Sovacool, B. K. (2013). Wind Turbines and Invisible Technology: Unarticulated Reasons for Local Opposition to Wind Energy. *Technology and Culture*, 54(4), 705-734. doi: 10.1353/tech.2013.0131
- Hoen, B., & Atkinson-Palombo, C. (2016). Wind Turbines, Amenities and Disamenitites: A study of Home Value Impacts in Densely Populated Massachusetts. *Journal of Real Estate Research*, 38(4), 473-504.
- Hoen, B., Brown, J., Jackson, T., Thayer, M., Wiser, R., & Cappers, P. (2015). Spatial Hedonic Analysis of the Effects of US Wind Energy Facilities on Surrounding Property Values. *The Journal of Real Estate Finance and Economics*, 51(1), 22-51. doi: 10.1007/s11146-014-9477-9
- Hoen, B., Wiser, R., Cappers, P., Thayer, M., & Sethi, G. (2011). Wind Energy Facilities and Residential Properties: The Effect of Proximity and View on Sales Prices. *Journal of Real Estate Research*, 33(3), 279-316.
- Huesca-Pérez, M. E., Sheinbaum-Pardo, C., & Köppel, J. (2016). Social implications of siting wind energy in a disadvantaged region – The case of the Isthmus of Tehuantepec, Mexico. *Renewable and Sustainable Energy Reviews, 58*, 952-965. doi: 10.1016/j.rser.2015.12.310
- Intergovernmental Panel on Climate Change (IPCC). (2011). *Special Report on Renewable Energy Sources and Climate Change Mitigation*. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Ipsos. (2010). Attitudes towards Energy Sources: A comparison of US and Global Attitudes *Ipsos Global Energy Barometer*.
- Jacquet, J. B. (2012). Landowner attitudes toward natural gas and wind farm development in northern Pennsylvania. *Energy Policy, 50*(Nov), 677-688.
- Jacquet, J. B. (2015). The Rise of "Private Participation" in the Planning of Energy Projects in the Rural United States. *Society & Natural Resources*, *28*(3), 231-245.
- Jacquet, J. B., & Stedman, R. C. (2013). Perceived Impacts from Wind Farm and Natural Gas Development in Northern Pennsylvania. *Rural Sociology*, 78(4), 450-472.
- Jami, A. A., & Walsh, P. R. (2014). The role of public participation in identifying stakeholder synergies in wind power project development: The case study of Ontario, Canada. *Renewable Energy, 68*, 194-202.
- Jami, A. A., & Walsh, P. R. (2017). From consultation to collaboration: A participatory framework for positive community engagement with wind energy projects in Ontario, Canada. *Energy Research & Social Science, 27*, 14-24.
- Janssen, S. A., Vos, H., Eisses, A. R., & Pedersen, E. (2011). A comparison between exposure-response relationships for wind turbine annoyance and annoyance due to other noise sources. *The Journal of the Acoustical Society of America*, *130*(6), 3746-3753. doi: 10.1121/1.3653984
- Jensen, C. U., Panduro, T. E., & Lundhede, T. H. (2014). The Vindication of Don Quijote: The impact of noise and visual pollution from wind turbines. *Land Economics*, *90*(4), 668-682.
- Kaliski, K., & Neeraj, G. (2013). *Prevalence of complaints related to wind turbines in northern New England.* Paper presented at the Meetings on Acoustics, ICA 2013, Montreal, Canada.
- Kempton, W., Firestone, J., Whitaker, P., Lilley, J., & Rouleau, T. (2005). The Offshore Wind Power Debate: Views from Cape Cod. *Coastal Management*, 33, 119-149.
- Klick, H., & Smith, E. (2010). Public understanding of and support for wind power in the United States. *Renewable Energy*, 35(7), 1585-1591.
- Knopper, L. D., & Ollson, C. A. (2011). Health effects and wind turbines: A review of the literature. *Environmental Health*, 10(1), 78.
- Knopper, L. D., Ollson, C. A., McCallum, L. C., Whitfield Aslund, M. L., Berger, R. G., Souweine, K., & McDaniel, M. (2014). Wind turbines and human health. *Front Public Health*, *2*, 63. doi: 10.3389/fpubh.2014.00063

- Krohn, S., & Damborg, S. (1999). On Public Attitudes Towards Wind Power. *Renewable Energy*, 16(1), 954-960.
- Krueger, A. D., Parsons, G. R., & Firestone, J. (2011). Valuing the visual disamenity of offshore wind power projects at varying distances from the shore: An application on the Delaware shoreline. *Land Economics*, 87(2), 268-283.
- Landry, C. E., Allen, T., Cherry, T., & Whitehead, J. C. (2012). Wind turbines and coastal recreation demand. *Resource and Energy Economics*, 34(1), 93-111.
- Lang, C., Opaluch, J. J., & Sfinarolakis, G. (2014). The Windy City: Property Value Impacts of Wind Turbines in an Urban Setting. *Energy Economics*, 44(2014), 413-421.
- Lantz, E., & Flowers, L. (2011). Social acceptance of wind energy projects: Country report of United States. *IEA Wind Task 28*.
- Larson, E. C., & Krannich, R. S. (2016). "A Great Idea, Just Not Near Me!" Understanding Public Attitudes About Renewable Energy Facilities. *Society & Natural Resources, 29*(12), 1436-1451. doi: 10.1080/08941920.2016.1150536
- Leiserowitz, A., Maibach, E., Roser-Renouf, C., Feinberg, G., & Rosenthal, S. (2014). Public Support for Climate and Energy Policies in November 2013 Yale University's Project on Climate Change Communication and George Mason University.
- Lilley, M. B., Firestone, J., & Kempton, W. (2010). The effect of wind power installations on coastal tourism. *Energies*, *3*(1), 1-22.
- Loss, S. R., Will, T., & Marra, P. P. (2013). Estimates of bird collision mortality at wind facilities in the contiguous United States. *Biological Conservation*, *168*, 201-209.
- Lutzeyer, S. (2013). *Essays in Offshore Wind Energy Development*. (Doctor of Philosophy Dissertation), North Carolina State University.
- Magari, S. R., Smith, C. E., Schiff, M., & Rohr, A. C. (2014). Evaluation of community response to wind turbine-related noise in Western New York State. *Noise and Health*, *16*(71), 228.
- Maruyama, Y., Nishikido, M., & Iida, T. (2007). The Rise of Community Wind Power in Japan: Enhanced Acceptance through Social Innovation. Energy Policy, 35(7), 2761-2769.
- McCunney, R. J., Mundt, K. A., Colby, W. D., Dobie, R., Kaliski, K., & Blais, M. (2014). Wind Turbines and Health: A Critical Review of the Scientific Literature. *Journal of Occupational and Environmental Medicine*, *56*(11), e108-e130.
- Michaud, D. S., Feder, K., Keith, S. E., Voicescu, S. A., Marro, L., Than, J., Guay, M., Denning, A., McGuire, D., Bower, T., Lavigne, E., Murray, B. J., Weiss, S. K., & van den Berg, F. (2016a). Exposure to wind turbine noise: Perceptual responses and reported health effects. *J Acoust Soc Am*, 139(3), 1443-1454. doi: 10.1121/1.4942391
- Michaud, D. S., Feder, K., Keith, S. E., Voicescu, S. A., Marro, L., Than, J., Guay, M., Denning, A., Murray, B. J., Weiss, S. K., Villeneuve, P. J., van den Berg, F., & Bower, T. (2016b). Effects of Wind Turbine Noise on Self-Reported and Objective Measures of Sleep. *Sleep*, *39*(1), 97-109. doi: 10.5665/sleep.5326
- Mulvaney, K. K., Woodson, P., & Prokopy, L. S. (2013a). Different Shades of Green: A Case Study of Support for Wind Farms in the Rural Midwest. *Environmental management*, *51*(5), 1012-1024.
- Mulvaney, K. K., Woodson, P., & Prokopy, L. S. (2013b). A tale of three counties: Understanding wind development in the rural Midwestern United States. *Energy Policy*, *56*(0), 322-330. doi: <u>http://dx.doi.org/10.1016/j.enpol.2012.12.064</u>
- National Academy of Sciences (NAS). (2007). *Environmental Impacts of Wind-Energy Projects*. Washington, DC: National Academy of Sciences.
- Nissenbaum, M. A., Aramini, J. J., & Hanning, C. D. (2012). Effects of Industrial Wind Turbine Noise on Sleep and Health. *Noise and Health*, *14*(60), 237.

- Olson-Hazboun, S. K., Krannich, R. S., & Robertson, P. G. (2016). Public views on renewable energy in the Rocky Mountain region of the United States: Distinct attitudes, exposure, and other key predictors of wind energy. *Energy Research & Social Science, 21*, 167-179. doi: 10.1016/j.erss.2016.07.002
- Paller, C. (2014). Exploring the Association between Proximity to Industrial Wind Turbines and Self-Reported Health Outcomes in Ontario, Canada. UWSpace. Retrieved from http://hdl.handle.net/10012/8268
- Pasqualetti, M. J. (2000). MORALITY, SPACE, AND THE POWER OF WIND-ENERGY LANDSCAPES. *Geographical Review*, *90*(3), 381-394.
- Pasqualetti, M. J. (2001). Wind Energy Landscapes: Society and Technology in the California Desert. Society & Natural Resources, 14(8), 689-699.
- Pasqualetti, M. J. (2011a). Opposing Wind Energy Landscapes: A Search for Common Cause. *Annals of the Association of American Geographers, 101*(4), 907-917. doi: 10.1080/00045608.2011.568879
- Pasqualetti, M. J. (2011b). Social barriers to renewable energy landscapes. *Geographical Review*, 101(2), 201-223.
- Pasqualetti, M. J., & Butler, E. (1987). Public Reaction to Wind Development in California. *International Journal of Ambient Energy*, 8(2), 83-90.
- Pasqualetti, M. J., Gipe, P., & Righter, R. W. (2002). *Wind Power in View: Energy Landscapes in a Crowded World* (M. J. Pasqualetti, P. Gipe & R. W. Righter Eds.). San Diego, CA: Academic Press.
- Pedersen, E., & Waye, K. P. (2004). Perception and Annoyance due to Wind Turbine Noise: A Dose-Response Relationship. *The Journal of the Acoustical Society of America*, *116*(6), 3460-3470.
- Petrova, M. A. (2013). NIMBYism revisited: public acceptance of wind energy in the United States. *Wiley Interdisciplinary Reviews: Climate Change, 4*(6), 575-601.
- Petrova, M. A. (2014). Sustainable Communities and Wind Energy Project Acceptance in Massachusetts. *Minnesota Journal of Law, Science & Technology, 15*, 529-689.
- Petrova, M. A. (2016). From NIMBY to acceptance: Toward a novel framework VESPA For organizing and interpreting community concerns. *Renewable Energy*, 86, 1280-1294. doi: 10.1016/j.renene.2015.09.047
- Phadke, R. (2010). Steel forests or smoke stacks: the politics of visualisation in the Cape Wind controversy. *Environmental politics*, *19*(1), 1-20.
- Phadke, R. (2011). Resisting and Reconciling Big Wind: Middle Landscape Politics in the New American West. *Antipode*, 43(3), 754-776. doi: 10.1111/j.1467-8330.2011.00881.x
- Phadke, R. (2013). Public Deliberation and the Geographies of Wind Justice. *Science as Culture, 22*(2), 247-255. doi: 10.1080/09505431.2013.786997
- Pohl, J., Hübner, G., & Mohs, A. (2012). Acceptance and stress effects of aircraft obstruction markings of wind turbines. *Energy Policy*, *50*(November), 592-600.
- Rule, T. A. (2014). *Solar, wind and land: Conflicts in renewable energy development*. New York, NY: Routledge.
- Shaw, K., Hill, S. D., Boyd, A. D., Monk, L., Reid, J., & Einsiedel, E. F. (2015). Conflicted or constructive? Exploring community responses to new energy developments in Canada. *Energy Research & Social Science*, 8, 41-51. doi: 10.1016/j.erss.2015.04.003
- Slattery, M. C., Johnson, B. L., Swofford, J. A., & Pasqualetti, M. J. (2012). The predominance of economic development in the support for large-scale wind farms in the U.S. Great Plains. *Renewable and Sustainable Energy Reviews*, *16*(6), 3690-3701. doi: <u>http://dx.doi.org/10.1016/j.rser.2012.03.016</u>
- Songsore, E., & Buzzelli, M. (2014). Social responses to wind energy development in Ontario: The influence of health risk perceptions and associated concerns. *Energy Policy*, *69*, 285-296.
- Songsore, E., & Buzzelli, M. (2015). Wind energy development in Ontario: A process/product paradox. *Local Environment, 20*(12), 1428-1451.

Sovacool, B. K. (2009). Rejecting renewables: The socio-technical impediments to renewable electricity in the United States. *Energy Policy*, *37*(11), 4500-4513. doi: 10.1016/j.enpol.2009.05.073

Sovacool, B. K. (2013). The avian benefits of wind energy: A 2009 update. *Renewable Energy*, 49, 19-24.

- Sovacool, B. K., & Ratan, P. L. (2012). Conceptualizing the acceptance of wind and solar electricity. *Renewable and Sustainable Energy Reviews, 16*(7), 5268-5279. doi: 10.1016/j.rser.2012.04.048
- Sullivan, R. G., Kirchler, L., Lahti, T., Roché, S., Beckman, K., Cantwell, B., & Richmond, P. (2012). *Wind turbine visibility and visual impact threshold distances in western landscapes.* Paper presented at the National Association of Environmental Professionals 37th Annual Conference, May.

Swofford, J., & Slattery, M. (2010). Public attitudes of wind energy in Texas: Local communities in close proximity to wind farms and their effect on decision-making. *Energy Policy, 38*(5), 2508-2519.

- Taebi, B. (2016). Bridging the Gap between Social Acceptance and Ethical Acceptability. *Risk Anal*. doi: 10.1111/risa.12734
- Thayer, R. L., & Freeman, C. N. (1987). Altamont: Public Perceptions of a Wind Energy Landscape. Landscape and Urban Planning, 14(1987), 379-398.
- Tiebout, C. M. (1956). A Pure Theory of Local Expenditures. *The Journal of Political Economy, 64*(5), 416-424.
- USDOE. (2015). Wind Vision: A New Era for Wind Power in the United States. Washington, DC: US Department of Energy.
- Vyas, C., & Hurst, D. (2013). Energy and Environment Consumer Survey: Consumer Attitudes and Awareness Toward 10 Smart Energy Concepts (pp. 35). Boulder, CO: Navigant.
- Walker, C., Baxter, J., Mason, S., Luginaah, I., & Ouellette, D. (2014a). Wind energy development and perceived real estate values in Ontario, Canada. *AIMS Energy*, *2*(4), 424-442.
- Walker, C., Baxter, J., & Ouellette, D. (2014b). Adding insult to injury: the development of psychosocial stress in Ontario Wind Turbine communities. *Social Science & Medicine*, 1-8.
- Walker, C., Baxter, J., & Ouellette, D. (2014c). Beyond rhetoric to understanding determinants of wind turbine support and conflict in two Ontario, Canada communities. *Environment and Planning A*, 46(3), 730-745.
- Warren, C., Lumsden, C., O'Dowd, S., & Birnie, R. (2005). Green on Green: Public Perceptions of Wind Power in Scotland and Ireland. *Journal of environmental planning and management, 48*(6), 853-875.
- Warren, C. R., & McFadyen, M. (2010). Does community ownership affect public attitudes to wind energy? A case study from south-west Scotland. Land Use Policy, 27(2), 204-213.
- Williams, W., & Whitcomb, R. (2007). *Cape Wind: Money, Celebrity, Class, Politics, and the Battle for Our Energy Future*: PublicAffairs.
- Wiser, R., & Bolinger, M. (2016). 2015 Wind Technologies Market Report. Berkeley, CA: Lawrence Berkeley National Laboratory.
- Wiser, R., Jenni, K., Seel, J., Baker, E., Hand, M., Lantz, E., & Smith, A. (2016). Expert elicitation survey on future wind energy costs. *Nature Energy*, *1*, 16135.
- Wolsink, M. (2000). Wind Power and the NIMBY-Myth: Institutional Capacity and the Limited Significance of Public Support. *Renewable Energy*, *21*(1), 49-64.
- Wolsink, M. (2006). Invalid theory impedes our understanding: a critique on the persistence of the language of NIMBY. *Transactions of the Institute of British Geographers, 31*(1), 85-91.
- Wolsink, M. (2007). Wind power implementation: The nature of public attitudes: Equity and fairness instead of 'backyard motives'. *Renewable and Sustainable Energy Reviews*, *11*(6), 1188-1207.
- Wolsink, M. (2012). Undesired reinforcement of harmful 'self-evident truths' concerning the implementation of wind power. *Energy Policy*, *48*, 83-87. doi: 10.1016/j.enpol.2012.06.010
- World Health Organization. (1999). *Guidelines for Community Noise*. (WA 776 99GU). Geneva, Switzerland: edited by Berglund, B., Lindvall, T., Schwela, D.H.

- World Health Organization. (2009). *Night Noise Guidelines for Europe*. (ISBN 978 92 890 4173 7). Copenhagen, Denmark.
- Wüstenhagen, R., Wolsink, M., & Burer, M. J. (2007). Social Acceptance of Renewable Energy Innovation: An Introduction to the Concept. *Energy Policy*, *35*, 2683-2691.
- Yoon, K., Gwak, D. Y., Seong, Y., Lee, S., Hong, J., & Lee, S. (2016). Effects of amplitude modulation on perception of wind turbine noise. *Journal of Mechanical Science and Technology*, *30*(10), 4503-4509.
- Zaunbrecher, B. S., & Ziefle, M. (2016). Integrating acceptance-relevant factors into wind power planning: A discussion. *Sustainable Cities and Society*, 27, 307-314. doi: 10.1016/j.scs.2016.08.018
- Zimmerling, J., Pomeroy, A., d'Entremont, M., & Francis, C. (2013). Canadian estimate of bird mortality due to collisions and direct habitat loss associated with wind turbine developments. *Avian Conservation and Ecology*, *8*(2).

Appendix A Table 1: Explanatory variables in N.A. wind acceptance literature & research suggestions

Explanatory Variable	Sub-variable	North American Citations	Further Research
Socio-economic aspects	Landowner compensation	Baxter et al., 2013 Firestone et al., 2012b Garcia et al., 2016 Gipe, 1995 Groth & Vogt, 2014 Jacquet, 2012 Walker et al., 2014b	What do hosting and non-hosting neighbors perceive to be the most appropriate forms of compensation? What forms of community-level compensation are preferred?
	Community investment & ownership models	Bolinger, 2005 Brannstrom et al., 2011 Fast et al., 2016 Ferguson-Martin & Hill, 2012 Songsore & Buzzelli, 2015 Sovacool & Ratan, 2012	More research is needed on this topic in the North American context. How to enable community investment in private developer- led wind projects?
	Property value impacts	Abbott, 2010 Fast et al., 2015 Firestone & Kempton, 2007 Heintzelman & Tuttle, 2012 Hoen et al., 2011; 2015 Hoen & Atkinson-Palombo, 2016 Walker et al., 2014a	What source of information is trusted for property value impacts? Are there community compensation mechanisms that could assuage concerns about property value impacts?
	Tourism impacts	Firestone et al., 2009 Landry et al., 2012 Lilley et al., 2010 Lutzeyer, 2013	Are there methods to enhance tourism near wind projects? Does this impact differ between onshore and offshore projects?
	Impacts on electricity rates	Baxter et al., 2013 Firestone & Kemption, 2007 Firestone et al., 2012a Petrova, 2016	
	Jobs and local economic development	Bidwell, 2013 Larson & Krannich, 2016 Mulvaney et al., 2013a; 2013b Olson-Hazboun et al., 2016 Slattery et al., 2012 Songsore & Buzzelli, 2015	How are local economic impacts perceived under more participatory development processes?
	Distributive justice / costs and benefits	Baxter et al., 2013 Brannstrom et al., 2011 Groth & Vogt, 2014 Huesca-Pérez et al., 2016 Hirsh & Sovacool, 2013 Jami & Walsh, 2017 Kempton et al., 2005 Larson & Krannich, 2016 Pasqualetti, 2011a Petrova, 2013; 2016 Phadke, 2011; 2013 Rule, 2014 Shaw et al., 2015 Slattery et al., 2012 Sovacool, 2009	What do developers and communities perceive as best practices for distributive justice? What are the preferred community compensation mechanisms to improve distributive justice? How are feelings of distributive justice influenced by participation in the planning process? How to enable community investment in private developer- led wind projects?

Sound Aspects	Health impacts	Baxter et al., 2013	What types of sounds cause the
		Knopper & Ollson, 2011 Knopper et al., 2014 Magari et al., 2014 Michaud et al., 2016a; 2016b Songsore & Buzzelli, 2014 Walker et al., 2014c	most stress or sleep disruption?
	Annoyance	Kaliski & Neeraj, 2013 Knopper & Ollson, 2011 Knopper et al., 2014 Nissenbaum et al., 2012 Petrova, 2016 Phadke, 2013	What types of sounds are most annoying? Is measured or modeled sound correlated to levels of annoyance? If not, how can the sound models be improved?
Visual & Landscape Aspects	Aesthetic aspects, beauty, & annoyance	Apostol et al., 2016 Brannstrom et al., 2011 Bush & Hoagland, 2016 Fast et al., 2015 Fergen & Jacquet, 2016 Firestone & Kemption, 2007 Firestone et al., 2015 Gipe, 1993; 1995 Hirsch & Sovacool, 2013 Jacquet & Stedman, 2013 Krueger et al., 2011 Larson & Krannich, 2016 Olson-Hazboun et al., 2016 Pasqualetti, 2000; 2001; 2011a Pasqualetti & Bultler, 1987 Pasqualetti et al., 2002 Petrova 2016 Phadke, 2010 Rule, 2014 Thayer & Freeman, 1987	What are the motivations for positive visual perceptions? Does physical geography influence attitudes or visual annoyance? Does the number of turbines or frequency of seeing turbines influence attitudes? How do visual simulations influence support/opposition? How can simulations, and how they are presented, be improved to better reflect actual developments?
	Economic effects of visual impacts (e.g. willingness to accept view) Place Attachment	Boatwright, 2013 Groothuis et al., 2008 Jensen et al., 2014 Krueger et al., 2011 Pasqualetti, 2001 Bidwell, 2013 Fast & Mabee, 2015 Firestone et al., 2009 Jacquet & Stedman, 2013	More research on place attachment is needed in the North American context.
Environmental concerns and attitudes	Environmental attitudes and perceived impacts	Abbott, 2010 Fergen & Jacquet, 2016 Firestone et al., 2009; 2012a Firestone & Kempton, 2007 Kempton et al., 2005 Larson & Krannich, 2016 Mulvaney et al., 2013a; 2013b Olson-Hazboun et al., 2016 Petrova, 2016 Thayer & Freeman, 1987	How do environmental concerns and motivations vary in different socio-policitical contexts?

	Climate change	Bidwell, 2015 Olson-Hazboun et al., 2016 Petrova, 2013; 2016 Slattery et al., 2012 Swofford & Slattery, 2010	Do residents believe the local wind project makes a difference with respect to climate change?
Planning process, fairness, trust	Community participation in development process	Bidwell, 2016b Bohn & Lant, 2009 Corscadden et al., 2012 Fast et al., 2016 Firestone et al., 2012b Groth & Vogt, 2014 Huesca-Pérez et al., 2016 Jacquet, 2015 Jami & Walsh, 2014; 2017 Phadke, 2011; 2013 Shaw et al., 2015 Songsore & Buzzelli, 2015 Sovacool & Ratan, 2012 Walker et al., 2014c	How should developers implement a more participatory, collaboartive planning process? What communication practices and techniques are most effective between stakeholders? What do project developers consider best practices?
	Fairness, trust, and relationships	Dwyer, 2016 Fast, 2015 Fast & Mabee, 2015 Fast et al., 2016 Firestone et al., 2012b Shaw et al., 2015 Songsore & Buzzelli, 2014	Do perceptions of fairness influence perceptions of economic impacts or reported health impacts?
Distance from turbines	Proximity hypothesis	Baxter et al., 2013 Groth & Vogt, 2014 Swofford & Slattery, 2010 Thayer & Freeman, 1987	Do residents move in and out over time based on attitudes (Tiebout sorting)? Do individuals living closest to turbines have distinct attitudes or impacts?
	Cumulative impacts	Olson-Hazboun et al., 2016 Petrova, 2013 Walker et al., 2014c	The influence of cumulative impacts on communities needs further study in North America
Other	Knowledge / information deficit	Baxter et al., 2013 Bidwell, 2016a Bosley & Bosley, 1988 Bush & Hoagland, 2016 Corscadden et al., 2012 Fast, 2015 Klick & Smith, 2010 Sovacool, 2009 Swofford & Slattery, 2010	What kinds of entities are trusted sources for information about wind energy? What are preferred methods for presenting information? How can a bi-lateral exchange of information between the hosts and developers be encouraged?
	Influence of local or federal policy on local acceptance	Fast & Mabee, 2015 Petrova, 2014 Songsore & Buzzelli, 2014; 2015	Compare policies that have increased acceptance to those that have decreased it
	Compare attitudes toward wind with other energy sources	Jacquet, 2012 Jacquet & Stedman, 2013 Shaw et al., 2015	More comparisons of preferences/attitudes toward different energy sources are needed.

		American wind acceptance literature North American Citations
Research Approach	Specific Method	
Quantitative Methods	One-time case study, statistical survey near existing wind facility	Baxter et al., 2013 Fergen & Jacquet, 2016 Firestone et al., 2015 Groth & Vogt, 2014 Jacquet, 2012; 2015
		Jacquet & Stedman, 2013 Magari et al., 2014 Mulvaney et al., 2013 Nissenbaum et al., 2012 Olson-Hazboun et al., 2016 Paller, 2014 Pasqualetti & Butler, 1987 Petrova, 2014; 2016 Slattery et al., 2012 Swofford & Slattery, 2010
	One-time case study, statistical survey; hypothetical wind facility	Thayer & Freeman, 1987 Bidwell, 2013; 2015; 2016a Boatwright, 2013 Corscadden et al., 2012 Firestone & Kempton, 2007 Firestone et al., 2009; 2012b Groothuis et al., 2008 Landry et al., 2012 Larson & Krannich, 2016 Lutzeyer, 2013
	Longitudinal case study survey (over time)	Firestone et al., 2012a
	Nationally representative opinion poll	Ipsos, 2010 Klick & Smith, 2009 Leiserowitz et al., 2014 Vyas & Hurst, 2013
	Economic modeling and choice experiments	Garcia et al., 2016 Groothuis et al., 2008 Heintzelman & Tuttle, 2012 Hoen et al., 2011; 2015 Hoen & Atkinson-Palombo, 2016 Kreuger et al., 2011 Landry et al., 2012 Lang et al., 2014 Lilley et al., 2010 Lutzeyer, 2013
	Epidemiological studies	Michaud et al., 2016a; 2016b Nissenbaum et al., 2012
Mixed Methods	Interview and Survey	Bosley & Bosley, 1988 Kreuger et al., 2011 Mulvaney, 2013a; 2013b Walker et al., 2014a; 2014b; 2014c
	Internations and Contract Analysis /	East at al 2015
	Interview and Content Analysis / Literature Review	Fast et al., 2015 Sovacool & Ratan, 2012

	[
Qualitative Methods	Literature review	Fast, 2015
		Fournis & Fortin, 2016
		Huesca-Pérez et al., 2016
		Knopper & Ollson, 2011
		Knopper et al., 2014
		Lantz & Flowers, 2011
		Pasqualetti, 2011b
		Petrova, 2013
		Rule, 2014
	Content analysis	Abbott, 2010
		Bohn & Lant, 2009
		Bolinger, 2005
		Ferguson-Martin & Hill, 2012
		Pasqualetti, 2011a
		Phadke, 2010
		Shaw et al., 2015
		Songsore & Buzzelli, 2014; 2015
	Interview / Focus Group	Dwyer, 2016
		Fast & Mabee, 2015
		Jami & Walsh, 2017
		Kempton et al., 2005
		Pasqualetti, 2001
		Phadke, 2013
		Shaw et al., 2015
		Sovacool, 2009
	Comment / Perspective /	Bidwell, 2016b
	Qualitative case study	Fast et al., 2016
	Quinter of the order of the day	Hirsch & Sovacool, 2013
		Jami & Walsh, 2014
		Pasqualetti, 2000
		Phadke, 2011
	Expert elicitation	Bush & Hoagland, 2016